Massachusetts Institute of Technology Handout 23 6.892: Theory of Parallel Systems April 3, 1997 Professor Charles E. Leiserson

Term Project Presentation

In addition to the proposal, we ask you to give a short presentation (3-4 minutes). The lecture of Thursday, April 10 will be dedicated to your presentations.

In the presentation you must briefly discuss the following:

ffl Description of the problem to be solved.

ffl Description of your proposed approach to the problem.

ffl Evidence for why you believe you will be successful doing your project in the time allocated.

Everybody in the class must give a presentation. If a project involves more than one person, all of them will have their 3-4 minutes slot. In this case, you must clearly explain how you divide the work and which part you will do.